Author’s response to reviews

Title: Assessment of strategies for switching patients from olanzapine to risperidone: a randomized, open-label, rater-blinded study

Authors:

Rohan Ganguli (gangulir@msx.upmc.edu)
Jaspreet S Brar (brarjs@upmc.edu)
Ramy Mahmoud (rmahmoud@ethus.jnj.com)
Sally A Berry (sberry@prdus.jnj.com)
Gahan J Pandina (gpandina@omjus.jnj.com)

Version: 2 Date: 13 May 2008

Author’s response to reviews:

Please note, this manuscript was accepted for publication in BMC Psychiatry, but the publishers requested that we transfer the accepted manuscript to BMC Medicine. Thus, no revisions have been made.